Author’s response to reviews

Title: The expression of TRM2A, a Novel Cell Cycle Regulated Protein, Identifies a Subset of Breast Cancer Patients with HER2 Over-expression that are at an Increased Risk of Recurrence

Authors:

David G Hicks (David_Hicks@urmc.rochester.edu)
Bagi R Janarthanan (Bagi.Janarthanan@roswellpark.org)
Ramya Vardarajan (Ramya.Vardarajan@RoswellPark.org)
Swati A Kulkarni (swati.kulkarni@roswellpark.org)
Thaer Khoury (Thaer.Khoury@roswellpark.org)
Daniel Dim (Daniel.Dim@roswellpark.org)
G. Thomas Budd (BUDDG@ccf.org)
Brian J Yoder (yoderbj@gmail.com)
Raymond R Tubbs (tubbsr@ccf.org)
Marshall T Schreeder (mts@ccihsv.com)
Noel C Estopinal (NRMCT@aol.com)
Rodney A Beck (rbeck@applied-genomics.com)
Yanling Wang (ywang@applied-genomics.com)
Brian Z Ring (bzing@applied-genomics.com)
Robert S Seitz (rseitz@applied-genomics.com)
Douglas T Ross (dross@applied-genomics.com)

Version: 6 Date: 10 February 2010

Author’s response to reviews:

Dear Sir/Madaam,

Please find submitted a revised version with formatting changes as requested. I hope that this is satisfactory.

Thank you, Doug